Author’s response to reviews

Title: Magnetic resonance enterography in pregnant women with Crohn's disease: case series and literature review

Authors:

Myriam D Stern (myriam.d.stern@gmail.com)
Uri Kopylov (ukopylov@gmail.com)
Shomron Ben-Horin (shomron.benhorin@gmail.com)
Sarah Apter (saraapter@gmail.com)
Marianne Michal Amitai (mamital@hotmail.com)

Version: 4 Date: 27 July 2014

Author’s response to reviews:

Dear editor,
Thank you for the positive view of our manuscript.
The following errors were corrected as requested:
#1. The name of the IRB is now stated#
#2. The patient’s age was removed from the table.
Sincerely,
Dr Uri Kopylov